Author's response to reviews

Title: beta hCG Secreting Non-Small Cell Lung Cancer: A Case Report

Authors:

Saakshi Khattri (saakshi_doc@hotmail.com)
Abhirami Vivekanandarajah (avivek27@gmail.com)
Seema Varma (seema_varma@siuh.edu)
Frank Kong (frank_kong@siuh.edu)

Version: 5 Date: 9 December 2010

Author's response to reviews:

The following changes have been made as requested-
1. Affiliations have been written in full as directed.
2. Term 'hypodensity' has been changed to 'metastasis'.
3. References have been mentioned in square brackets instead of superscript form.
4. Figures 6 and 7 have been included in the text.
5. Email addresses have been mentioned in the format desired.
6. Section heading has been changed from 'Background' to 'Introduction'.